Author’s response to reviews

Title: The beneficial use of ultramicronized palmitoylethanolamide as add-on therapy to Tapentadol in the treatment of low back pain: a pilot study comparing prospective and retrospective observational arms

Authors:

Maria Passavanti (beatrice.passavanti@libero.it)
Pasquale Sansone (pasquale.sansone@unicampania.it)
Caterina Aurilio (caterina.aurilio@unicampania.it)
Vincenzo Pota (vincenzo.pota@unicampania.it)
Daniela Fierro (danielafierroh@gmail.com)
Marco Fiore (marco.fiore@hotmail.it)
Maria Pace (caterina.pace@libero.it)

Version: 1 Date: 06 Jul 2017

Author’s response to reviews:

Dear Editor,

we provided the informations requested.

We hope that the revised version meets the standards of your prestigious journal.

Sincerely,

Marco Fiore, MD